



Article Challenges for Energy Transition in Poverty-Ridden Regions—The Case of Rural Mixteca, Mexico

Laura-Patricia Oviedo-Toral *, Davi Ezequiel François and Witold-Roger Poganietz

Institute for Technology Assessment and System Analysis (ITAS), Karlsruhe Institute of Technology (KIT), 76133 Karlsruhe, Germany; davi.francois@kit.edu (D.E.F.); witold-roger.poganietz@kit.edu (W.-R.P.) * Correspondence: patricia.oviedo@kit.edu

Abstract: This paper presents distinct scenario pathways and their storylines resulting from an analysis of interdependencies. We identified the main drivers of a proposed renewable energy transition in rural Mixteca-Puebla, Mexico. By analyzing the main factors involved in alleviating impoverished communities in the rural region, we show the varying degrees to which these drivers influence, support, or hinder a promising energy transition. A Cross-Impact Balance Analysis was conducted to explore the multiple inter-relationships among a set of conditions. This methodology allowed us to evaluate the relationships between social, political, cultural, and environmental variables. The main drivers were identified as clusters of several elements, in which the *uncertainties in governance* and the *legal system* trigger the inter-relationship of forces in the area. The focus of this paper is to show how the societal aspects affect the structural energy transformation and its capacity for adaptation in future trends envisioned for the area. This research contributes to the use of technological transformation as a means to alleviate poverty in a rural area. These outcomes give insights regarding the conditions to be considered, in respect to possible–encouraging, but also pessimistic pathways for the region in coming decades.

Keywords: scenario analysis; cross-impact balance; energy transition; rural areas; socio-technical system; renewable energy; transformation pathways; poverty alleviation

1. Introduction

The imminent threat of climate change and its impact on the human habitat demands an energy system transition from fossil fuels toward renewable sources. Many countries, cities, and regions have envisioned how to become low carbon communities, but still lack a defined framework including potential unintended outcomes, e.g., new forms of property and digital knowledge [1]. The energy transition can frame the shift in a way that inadvertently downplays the profound economic, social, political, and cultural disruptions that such energy changes entail. Moreover, a sustainable energy system transition involves multiple factors besides the technological transformation [2]. Literature on sustainability transitions addresses the deployment of specific technologies [3,4], or focuses on financial aspects [5] or environmental concerns [6]. Social factors are usually taken as a constant in the analysis [7], and the complex multidisciplinary processes that sustain or hinder social features [8] are rarely analyzed. Thus, there is a need to explore energy transition processes through the shifts between the dynamics of networks, communities, and governance contexts in which the different types of actors interact [9].

While the concept of "energy transition" has a connotation related to security, efficiency and sustainability among the high-income economies [10,11], in low-income countries the "energy transition" deals with the dilemma of gaining access to affordable energy services, without becoming trapped in a fossil fuel-intensive future [12]. Low-income economies face the challenge of sustainable development to promote energy justice and equity, economic development and poverty alleviation, whilst contributing to curbing climate change [13]. A renewable energy transition in rural areas of low-income countries, where



Citation: Oviedo-Toral, L.-P.; François, D.E.; Poganietz, W.-R. Challenges for Energy Transition in Poverty-Ridden Regions—The Case of Rural Mixteca, Mexico. *Energies* 2021, 14, 2596. https://doi.org/ 10.3390/en14092596

Academic Editor: Vincenzo Bianco

Received: 28 March 2021 Accepted: 28 April 2021 Published: 1 May 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the affordability of electricity provides limited energy availability to its population, could offer an opportunity to mitigate poverty. Nevertheless, in many cases around the world, large energy projects are carried out without paying attention to the potential adverse impacts on poor people [14–16]. The provision of clean energy, as Burke [17] suggests, may also provide an improvement not only to basic services, but in job opportunities, additional sources of income, value added to renewable energy sources, and reductions in climate change impacts, among others.

Extensive research has established and measured energy transition success through its choice and promotion of a specific technology [3–5,11,12,17–21]. However, the suitable integration of societal needs has not been sufficiently considered [22]. Research on energy transitions in low-income economies has mainly been focused on choosing the appropriate (renewable) energy technology and its expansion, aiming mostly to tackle energy poverty [23]. However, access to energy does not necessarily alleviate income poverty or any other dimensions of it [24,25] when discussing low-income economies. Hence, identifying context-specific factors critical to the success of poverty alleviation research is vital. Sovacool recognized that when the energy technology system considers the societal context, renewable energy systems can be effective [20]. Recent research suggests that the energy transition should not be seen as a transfer of technologies, but rather as a transformation of the entire system [26]. Miller calls a system that delivers social value [27] and incorporates the social dynamics of diverse communities into its design a "socio-energy system" [28]. The Poor People's Energy Outlook argues that it becomes possible to escape the vicious cycle of poverty when people living in poverty have the sustainable energy access they need to grow enterprise activities, both small and large [22]. Multiple strategies have been designed in the aim to reduce poverty: social assistance payments [29] as a temporary alleviation of vulnerability; development of synergies in value-chain sectors as agriculture [30]; and supporting income-generating activities through the access of microfinance activities [31]. However, the drawback of these programs is that although they tackle the specific dimensions of poverty for which the program has been designed, they do not reflect on cultural or any social aspects. Moreover, there is no conclusive evidence regarding an improvement in other dimensions of poverty, such as health, education, or livelihood [32].

The evaluation of future systems analysis includes scenario simulation, because this offers a systematic approach about the future. Some studies have analyzed the energy transformation through the economic lens, while other uncertainties have been set aside, i.e., political or cultural concerns [18,33]. In this sense, methods such as scenario analysis can help to deal with uncertainty in societies. Model-based scenario analyses are practical tools concerning possible alternative futures [34]. Different types of scenario techniques have been used to evaluate future developments [35], most of them based on intuitive logic. An approach that describes the context in a more systematic way through the analysis of its interdependent factors is the Cross-Impact Balance (CIB) analysis [36]. An argument in favor of CIB includes the coupling of storylines in the model as a means to improve analysis and facilitate communication among parties [37]. Storylines are the qualitative narratives that describe the main trends in socio-economic, political, technological, and environmental drivers of change and their inter-relationships [38].

Despite the importance of societal needs, a thorough understanding of how to integrate these into the technological transformation is still necessary [17]. Literature is scarce, particularly with regard to the emerging role of the paradigm shift, in which technology is not the end of the transformation [35] but the means of the transition. Frequently, energy transformation is seen primarily as a change of the required energy technologies without considering that energy systems are embedded in societal dynamics, i.e., the transformation should be understood as a socio-technical challenge [39]. The presented research considers the energy transition as the means of a major social reconfiguration to contribute to economic development and poverty alleviation in rural regions. The novelty of this research is the shift of this perspective, which is the result of an intensive discussion in the European and North American literature [40], to the challenges of a poverty-ridden region in the Global South. By taking the case of rural Mixteca, we sought to understand the energy transformation as the means to improve living conditions and pursue the alleviation of poverty. This paper explores the possibility of taking advantage of the energy transition to contribute to social and economic development in rural regions.

The overarching aims of this paper are threefold: (1) to analyze the challenges of transforming a poverty-ridden area through a renewable energy transition, considering societal aspects as variables; (2) to show a series of plausible future outcomes (scenarios) derived from the inter-relationship of impacts between the drivers of change (descriptors); and (3) to analyze the different scenario pathways and storylines resulting from these interdependencies. The focus of the paper is how societal aspects affect the structural energy transformation and its capacity of adaptation in the future trends envisioned for the area. To this end, we used scenario analysis through the Cross-Impact Balance method, because it offers accountability for future uncertainties regarding how the impact factors evolve over time. This approach integrates the analysis of the complex interactions of the political, economic, technological, and social correspondent factors of the energy system.

Section 2 of this paper introduces Mixteca and the current situation. Section 3 discusses the methodology used: Cross-Impact Balance, the descriptors obtained, and the different scenarios resulting for our case. Section 4 describes the two clusters and the two storylines behind the scenarios obtained in Section 3. Section 5 provides a discussion of the findings where the main drivers and a hypothesis of the contrasting outcomes are addressed. To conclude, some final comments are provided in Section 6.

2. Mixteca-Puebla: Characterization of the Area

The Mixteca region includes parts of the Mexican states Guerrero, Oaxaca, and Puebla. This research focuses solely on the Mixteca region in the state Puebla. Mixteca in Puebla is located in the southwest of the state (see Figure 1). The region covers an area of 11,025 km², representing 32.5% of the state's territory, with 45 municipalities conforming the region. Its population of 254,100 inhabitants is scattered across 472 locations, representing 4.5% of the total state population; Mixteca is essentially considered a rural region [41].



Figure 1. Location of Mixteca region in the state of Puebla, Mexico.

The landscape is arid, semi dry, and hot, with a steady temperature of 25 °C. The average solar radiation index for the whole state of Puebla is 5.5 kWh/m^2 [42], while in the Mixteca region values as high as 6.4 kWh/m^2 have been reported [43].

The main economic activities are limited to subsistence farming, goat breeding, and palm weaving.

The region is poverty-ridden. About 79% of the population lives below the income poverty line [44]. Furthermore, they are deprived in at least one of the social dimensions which constitute the multidimensional concept of poverty [24,45]. These social dimensions are income, education, health services, social security, food, housing and quality space, basic services, social cohesion, and accessibility to paved roads [46]. About 30% of the population under the income poverty line live under extreme poverty conditions. Their income is below the cost of the basic food basket (extreme income poverty line) and is accompanied by three or more social deprivations. A chart showing the trend in poverty and vulnerability over the past decade is presented in Figure 2.





Characteristics of the poverty in Mixteca are highly relevant to structural deficiencies related to the precariousness of housing, its materials, and lack of basic services. Basic services refers to basic sanitation—to which 77% of the population has access, water supply near to their homes 76%, and electricity access 93% [47]. Only 54% of the inhabitants are covered by all three basic services. In addition to the low access to basic infrastructure, the quality of the provided services is meagre. The water service quality is under intermittent conditions; electricity, currently supplied through fossil fuel generation, is delivered under a poor technical and commercial efficiency [48], which makes it unviable to use for productive purposes. Lack of access to electricity is acknowledged as a sign of marginalization and vulnerability [24,49].

The lack or inadequate provision of basic services, in particular of energy, has a major impact on women, who are typically responsible for collecting and managing traditional sources of fuel(biomass). Additionally, because most men, husbands and brothers, emigrate, women are becoming in charge of households and communities, partly impeding the required empowerment of women. Alleviating time spent on activities that could be avoided with modern technology could relieve the burden and women could engage in more productive activities.

Mexico's education system is relatively weak, despite significant public investment in the sector. Children in Mixteca face unequal education opportunities, and the quality of education services that reach these communities remains low. Consequently, the average level of education in the area is less than 6 years. The vast majority of schools lack facilities such as laboratories, libraries, or sport areas; more strikingly, they lack basic services. Figures for Mixteca are not available, but those for the entire state of Puebla show that 18.5% of public schools do not have toilet facilities, 28.2% do not have electricity, and 27.8% do not have running water [50]. Given the deprived conditions of Mixteca, the percentages for the area should be higher, because 80% of the community primary schools are located in populations with a high or very high degree of marginalization [51].

Mixteca inhabitants have emigrated over several decades in search of work and a source of secure income. The preferred destination is north, mainly the United States, where growing demand for unskilled labor, mainly in agriculture, started a trend several decades ago. Mixteca emigrants are more likely to be males of working age, from 18 to 64 years old, which is the reason why the proportion of women living in Mixteca is on average 20% higher than men; reaching 25% in the range from 25 to 30 years old. The share of females under 18 years old is 49% [52]. Emigrants usually leave women behind who then become responsible for household and community decisions; therefore, women can gain a non-intended partial empowerment.

As a result of emigration, remittances are an important income source, manifesting a high degree of economic dependence not only for families, but also for the whole of rural Mixteca. Remittances have increased over the past years, providing support for private consumption, particularly of low-income families (Figure 3). In 2018, Mixteca contributed 13% of the total received remittances in the state of Puebla, although only 4.5% of the entire population of the state lives in Mixteca. Remittances predominantly are used as part of an income generation strategy for households rather than to stimulate a reduction in labor supply. The typical role of emigrants is capital delivery to their communities of origin.



Figure 3. Inflow of remittances to Mixteca and share of state participation, authors own calculations based on data from [53].

According to the Organisation for Economic Co-operation and Development (OECD) data, Mexico has the lowest financial inclusion among its members [54]; only one-third of the population has access to a savings account in the country, not to mention credit. According to estimates, only 6% of the population in Mixteca has knowledge or access to financial services [55]. The lack of financial education in rural areas is another aspect to consider, because people still believe that they do not need to learn to manage their limited budgets [56]. Thus, inhabitants tend to favor informal ways of financial societies lack rigorous standards. Moreover, there are no official banking institutions in the area, except for one bank located in the most developed community.

The weak rule of law is present in Mixteca. For example, electoral manipulation and vote-buying are ubiquitous, in exchange for future benefits such as federal social programs. One in two people in Mexico was offered a bribe for their vote, and one in four was threatened with retaliation [57]. The issue is exacerbated in Mixteca by factors such as the population's financial dependence and the association with low levels of schooling [58], which are well-known and widespread. In transgressions related to corruption, impunity reached 98% [59]; simple acts such as requests for public services, among other interactions with the government, are actions that involve corruption and which are also experienced among the poorest communities [60].

As mentioned above, the region shows a high solar radiation, and thus good conditions for solar power plants. However, currently, the Mexican state has decided to support fossil fuel infrastructure at the expense of current and future renewable energy investment [61,62]. This includes the construction of a new oil refinery and a new budget allocation to the modernization of coal, diesel, gas, and oil-fueled power plants. The decision to favor fossil fuel generation over renewable energy now positions Mexico on a path that hinders renewable energy generation, risking projects under development. This decision, limiting Mixteca's future plans on supporting renewable energy projects, could restrict its future development of clean energy [63].

3. Methodology: Cross-Impact Balance Analysis

The overall idea of the Cross-Impact Balance (CIB) analysis is to generate plausible context energy scenarios; in our case, of the energy system in rural Mixteca, which address not only techno-economic variables, but also societal non-quantitative variables, such as culture, politics, or the environment (i.e., the overall socio-economic, political, cultural, and environmental context in which energy systems are embedded) [36].

The selected CIB approach offers useful advantages for our purposes. Its qualitative orientation with respect to judgments and evaluation procedures meets the typology of data we face in this research; it balances logic with a theoretical basis of the system. The approach has been successfully applied in diverse and multiple research fields such as waste [64], water [65], politics [66], education [67], health [68], mobility and transport [69], and energy [70] among others; for further comparisons, please refer to Weimer-Jehle [71]. The CIB approach is implemented in four steps [34,37]:

(1) Defining the context. A selection of descriptors, which characterize the energy system of Mixteca, needed to be defined and understood as a socio-technical system. The selected descriptors represent social and cultural aspects, i.e., on emigration, ethnic identification, education, community organization, and women's empowerment, and political features, such as governance and uncertainties, policies, and the legal system. Furthermore, economic facets consisting of income, wealth distribution, and financial markets are addressed. Environmental factors such as climate change and its impacts on the population are also included under the model input data (Table 1).

This was achieved through interviews with fourteen experts in Mixteca and in three other Federal States of Mexico, as well as through a literature search. The experts had solid experience in rural development, sociology, energy research, technology assessment and policy, and the panel was formed by members of recognized affiliations such as CONACYT (Consejo Nacional de Ciencia y Tecnología—Mexican Council of Science and Technology), Mexican scientific thematic networks, Non-Governmental Organisations (NGOs) with a local presence in Mixteca (seven persons), and governmental institution members (five persons). In addition, five communities in Mixteca were visited, where nineteen families were interviewed. The detailed definition of each descriptor and its states can be found under the Supplementary Material, Table S1: Descriptors and states.

(2) Identifying the future system-states. To address possible trends and uncertainties, a set of two to four alternative future states were defined and assigned to each descriptor. These future states were selected through the group of experts involved. The selected descriptors and their alternative futures for the region under review are summarized in Table 1.

Deceminator

(3) Identifying the interdependencies and building up the cross-impact matrix (CIM). The inter-relationships between descriptors were valued using an integer, ranging, in our case, from -3 to +3, with -3 indicating a strong trade-off relationship, whereas +3 indicated a strong supporting relationship. The quantification of the interdependencies was performed with the assistance of the experts. An example of this evaluation is shown in Figure 4.

The outcome of the quantification of all interdependencies between system states is a CIM, which is shown in Appendix A, Figure A1.

(4) Identification and analysis of the scenarios. Using ScenarioWizard v4.31 (it can be downloaded from: www.cross-impact.org), consistent combinations of descriptorstates were identified. Each consistent combination of all descriptors described a scenario (Figure 5). The scenarios were analyzed to identify relevant driving forces and the political, societal, economic, and technological conditions of possible future developments (see Section 4).



Figure 4. Exemplary network inter-relationship between two descriptors and its states. It shows the influence (promotes: + or hinders: – and its degree (here: $-2 \dots 2$). Descriptor G can exert influence on descriptor T (active), and at the same time descriptor G can be influenced by descriptor T (passive).

Clate 2

Chata A

State 1	State 2	State 5	State 4		
A1 Return emigration	A2 Permanent emigration with bond	A3 Permanent emigration without bond			
B1 Low ethnic identification	B2 High ethnic identification	B3 Pluricultural			
C1 Less than 5 years	C2 From 5 to 9 years	C3 More than 9 years			
D1 Labor	D2 Remittances	D3 Remittances plus labor			
E1 No access to any service	E2 Partial access to services including water	E3 Partial access to services including electricity	E4 Access to all services		
F1 Poor community organization	F2 Limited to labor	F3 Support includes economic contribution			
G1 Stable job and min. or below min. wage	G2 Stable job above minimum wage	G3 No secure job and below minimum wage	G4 No secure job and above minimum wage		
H1 Low uncertainties	H2 Strong uncertainties with growth	H3 Strong uncertainties without growth			
I1 Restrictive policies on new energy systems	I2 Supportive policies on new energy systems	C C			
J1 Low investment or none	J2 High level of investment				
M1 Inexistent or low	M2 Existent or good	M3 Excellent			
N1 Inexistent or very low	N2 Existent or good				
	A1 Return emigration B1 Low ethnic identification C1 Less than 5 years D1 Labor E1 No access to any service F1 Poor community organization G1 Stable job and min. or below min. wage H1 Low uncertainties I1 Restrictive policies on new energy systems J1 Low investment or none M1 Inexistent or low N1 Inexistent or very low	A1 Return emigrationA2 Permanent emigration with bond B1 Low ethnic identificationB1 Low ethnic identificationB2 High ethnic identificationC1 Less than 5 years D1 LaborC2 From 5 to 9 years D2 RemittancesE1 No access to any serviceE2 Partial access to services including waterF1 Poor community organizationF2 Limited to laborG1 Stable job and min. or below min. wageG2 Stable job above minimum wage H1 Low uncertainties I1 Restrictive policies on new energy systemsJ1 Low investment or noneJ2 Kiph level of investmentM1 Inexistent or lowM2 Existent or goodN1 Inexistent or very lowN2 Existent or good	A1 Return emigrationA2 Permanent emigration with bond B2 High ethnic identificationA3 Permanent emigration without bond B3 PluriculturalG1 Less than 5 yearsC2 From 5 to 9 years D2 RemittancesC3 More than 9 years D3 Remittances plus labor E3 Partial access to service services including waterC3 More than 9 years D3 Remittances plus labor E3 Partial access to services including electricityF1 Poor community organizationF2 Limited to laborF3 Support includes economic contributionG1 Stable job and min. or below min. wage H1 Low uncertainties I1 Restrictive policies on 		

Table 1. Descriptors and states.

State 2

Ciata 1

Descriptor	State 1	State 2	State 3	State 4
O. Financial market in rural economy	O1 Limited access to formal financial market	O2 Access to informal financial market	O3 No access to formal or informal market	
P. Legal System	P1 Law enforcement	P2 Aggravate	P3 Not effectively enforced	
Q. Climate change R. Environmental effects on population	Q1 High impact	Q2 Low impact		
	R1 High impact	R2 Low impact		
S. Women's empowerment	S1 Limited or no empowerment	S2 Full attained	S3 Partial	
T. Community organization	T1 Poor community organization	T2 Limited to labor	T3 Support includes economic contribution	

Scenario No. 1	Scenario No. 2	Scenario No. 3	Scenario No. 5	Scenario No. 4	Scenario No. 6	Scenario No. 7	Scenario No. 8				
A. Emigration:		A. Emigration:		A Emigration:							
A2 Perm w/bond	A3 Permar	nent emigration w	ithout bond	A1 Return emigration							
B. Ethnic identity:		B. Ethnic identity:			B. Ethnic	identity:					
B2 High		B3 Pluricultural			B2 High ethnic	identification					
	C. Edu	ication:			C. Educ	cation:					
	C1 Less th	nan 5 years			C2 From 5	to 9 years					
D. Source	e income:	D. Source	e income:		D. Source	income:					
D2 Rem	ittances	D3 Remittand	ces plus labor	D1 Labor							
E. Basic s	services access (water, electricity, c	Irainage):		E. Basic Services:		E. Basic				
	E1 No access	to anv service		E3 Partial acces	ss to services inclu	udina electricity	F4 All				
F. Population a	cceptance of ren	w energy plans an	d participation:	F. Population a	cceptance of renw	energy plans ar	d participation:				
	F2 Limite	d to labor		F3 Econ.contrib.	F2 Ltd to labor	F3 Economi	c contribution				
	G. Job an	d earning:			G. Job and	earning:					
G3 N	o secure job and	below minimum	wage		G2 Stable job abov	e minimum wag	е				
	H. Governance	e uncertainties:		H. Governance uncertainties:							
H	3 Strong uncertai	nties without grow	rth	H1 Low uncertainties							
I. Governr	mental policies fo	r integrated energ	ysystem:	I. Governr	mental policies for	integrated energ	ysystem:				
I1 Re	strictive policies o	on new energysys	tems	l2 Sup	oportive policies or	n new energysys	tems				
	J. Investments or	n energy research	:		J. Investments on	energy research					
	J1 Low inves	tment or none			J2 High level o	ofinvestment					
M. Cooperation	n between goverr	nment, private inve	estors, NGOs :	M. Cooperatio	n between governi	ment, private inve	estors, NGOs :				
	M1 Inexis	tent or low			M3 Exc	ellent					
N. Added Va	lue creation from	the renewable er	ergy sector:	N. Added Value creation from the renewable energy sector:							
	N1 Inexister	nt or very low		N2 Existent or good							
O. Financi	al market in rural	economy:		O. Financial market in rural economy:							
O3 No acces	s to formal or info	ormal market		O2 Access to informal financial market							
	P. Legal	System:		P. Legal System:							
	P2 Age	gravate		P1 Law enforcement							
	Q. Climat	e change:		Q. Climate change:							
	Q1 Higl	n impact		Q2 Low impact							
R.	Environmental e	ffects on population	on:	R. Environmental effects on population:							
	R1 Higl	n impact		R2 Low impact							
S. Wo	omen's empower	ment:	S. Women empow:	S. Women empow:	S. Women's er	npowerment:	S. Women empow:				
S1 Lim	ited or no empow	rement	S3 Partial	S2 Full attained	S3 Pa	rtial	S2 Full attained				
		T. Co	ommunity organiz	zation: T. Comm.							
		-	T2 Limited to labo	organiz or T3 Econ contrib							

Figure 5. Consistent scenarios identified via CIB.

Table 1. Cont.

4. Storylines

4.1. Driving Forces

In order to select the role of the descriptors in our system, an evaluation of their impacts is a helpful way to consider the driving forces. Once we plotted all impact values, whether active (*y*-coordinate) or passive (*x*-coordinate), in a chart, we obtained the system grid shown in Figure 6. Active sum accounts for the number of descriptors, which are influenced by another descriptor. Passive sum shows the number of descriptors, which affect another selected descriptor. A high active sum and a comparable low passive sum indicates a driving force; in Figure 6, they are situated at the top-left of the chart. Descriptors with a low active but high passive sum are those which are more reactive to changes of the system than actively influencing it. They are mostly situated in the lower-left part of the chart. A third category represents those descriptors with no large discrepancies between active and passive sum; they influence a considerable part of the system under review, but they are also very much influenced by other descriptors. Weimer-Jehle refers to this type of descriptors as usually connected with the potential emergence of complex system behavior [71].



Figure 6. Active–passive positions of the descriptors.

Considering this structuring of the descriptors, the main driving forces are *P. Legal system* and *H. Governance uncertainties*, which exert more control than those at the right or bottom. In addition, as a third driving force, we identified *Q. Climate change*. This last descriptor has a special position in the system because, on the one hand, it is influenced by the behavior of the system; on the other hand, the development of the descriptor is also determined by factors outside of the system under review.

Highly connected descriptors are M. Cooperation between government, private investors, NGOs; I. Governmental policies for integrated energy system; N. Added Value creation from the renewable energy sector; C. Education; and J. Investments on energy research.

The rest of the descriptors show generally rather low active sums, but comparable high passive sums. The tight inter-relationship among all the descriptors is an intricate web, as shown in Figure 7, depicting the general complexity of the system under review. The following Sections 4.3–4.5 present an in-depth analysis of this complexity.



Figure 7. Interdependencies among descriptors. The arrows indicate the direction of the influences.

4.2. Characterization of Clusters

As mentioned above, a total of eight consistent scenarios were identified (Figure 5). These eight scenarios have been arranged into two clusters, because they shape very different outcomes/future developments.

The synopsis under cluster 1 consists of scenarios 1, 2, 3 and 5, whose storyline "Back to the XIX century" explores the worsening of social, economic, and political conditions in Mixteca. The cluster describes a situation which is characterized by an aggravated legal system with a strong uncertainty regarding governance and no economic growth perspectives. Additionally, the region is highly affected by climate change. The combination of these driving forces depresses the economic situation of the population, reducing the incentives to invest in education and decent jobs in the region, and promotes emigration as unskilled workers. A lack of access to formal financial markets also hinders investment in decent jobs. The bad economic situation also impedes investment in basic services, further deteriorating living standards and the prospects of remaining in the region. With respect to the energy system, the traditional orientation of the energy policies prevails, thus discouraging investments in renewable energy sources, as well as in the participation of civil society.

These conditions are comparable to those experienced by past generations during the early 19th century, where the country started its transition as an independent society, characterized by widespread poverty and lack of opportunities in rural areas.

Synopsis 2 consists of the cluster of scenarios 4, 6, 7 and 8, whose storyline "Hope for a better future" contrasts with synopsis 1. It envisions a future with an improvement of social, economic, and political conditions, where the low uncertainties in governance combined with law enforcement in the legal system provide an improved framework, with stable sources of employment and income generation in the region. The support of renewable energy policies for the energy transition increases the potential of investment

in these projects, not only from cooperation with entities outside the region, but from the population itself, who will be supportive and willing to self-finance its projects, aiming for self-sufficiency in the energy supply. Therefore, contrasting with synopsis 1, the self-generation of jobs and income within the communities will be a milestone aspired for and achieved, thus avoiding emigration, improving the chance of education, and increasing the possibility of lower impacts from climate change with lower impact on the population. Hence, a likely improvement in the quality of life would reduce poverty in the hope for a better future.

The overall appreciation is that cluster 1 differs from cluster 2 in a divergent pathway; the main drivers such as legal system and governance uncertainties produce opposite outcomes that lead to either hindering or promoting appropriate development conditions in the area.

4.3. Cluster 1, Synopsis: "Back to the XIX Century"

Synopsis 1 is characterized by the aggravation of the legal system, whose ineffective judicial system (descriptor P) will affect the region. Corruption and impunity will undermine the rule of law, a situation comparable to the status quo in Mixteca. Using Mexican data, because specific data for Mixteca are not available, on a scale of 0 (high corruption) to 100 (no corruption), Mexico achieves just 29, i.e., it ranks 130 out of 198 analyzed countries [72]. In the context of rampant corruption, impunity and the weak rule of law, the security crisis, and the aggravation of the legal system (state P2), development in Mixteca will be a tough challenge. Another dominant driver in this synopsis is the strong uncertainty in governance (descriptor H). Conflicting policies, programs, and communication between national and regional levels will contribute to increased uncertainties. Insufficient state capacities, both geographically and across policy sectors, will presumably undermine the effective and coherent implementation of policies. As a result, the energy sector will be exposed to hazards; the transition from a fossil fuel economy to the use of renewable energy sources is a tough future under synopsis 1. The future under this scenario challenges economic growth, affecting the weak economy in Mixteca.

The worsening of the legal system, combined with strong governance uncertainties, would exert a powerful negative influence on adopting local policies to support renewable energy systems in the region (descriptor I). It will reverse the renewable path the country had envisioned and to which they had committed in 2012, significantly hindering the integration of local renewable energy projects and restricting its future development (state I1). Therefore, scarce investment in research into renewable and clean energy will be the future trend, and Mixteca will not be able to profit (descriptor J). Under these circumstances, investments in research and development of renewable energy will not be a priority (state J1); on the contrary, Mixteca will depend on fossil fuels in future.

Due to the stagnation of renewable energy policies, the future value added of solar energy (descriptor N) will be not considered (state N1) under this future trend. Notwithstanding the high radiation levels in the region, the lack of support for renewable energy policies would provide unequal conditions to add value with clean energy projects. This situation would lead to low interest and participation from the population (descriptor F), limited to providing a work force (state F2) in the rural area to contribute to renewable energy aspects. Thus, societal and economic conditions could inhibit the technological transition in the territory.

Prioritizing fossil fuels will also undermine the cooperation (descriptor M) between government, private investors, and NGOs on renewable energy projects, discouraging investments due to the meager value added (state M1). The decision to favor fossil fuel generation over renewable energy will also put Mixteca on a path that is even more inconsistent with mitigation measures to avoid strong impacts of climate change. Under the conditions shown in synopsis 1, Mixteca will be highly vulnerable to the impacts of climate change (state Q1), in the form of more extreme weather patterns such as rising temperature, heat waves, unusual rain seasons, and acute and longer droughts [73], which will unavoidably aggravate existing social and economic inequalities.

These adverse conditions from climate change will severely affect the low productivity from agricultural jobs, tending livestock, hauling water and processing agricultural products. It will also promote a high impact on population (descriptor R, state R1) who would seek to leave the area, mainly outside of the country, as a way to overcome the intensified poverty in Mixteca. Under these circumstances, the trend G3 "no secure job and below minimum wage" is a consequential outcome. The uneven distribution of income is highlighted in low-skilled and rural Mixteca, which is one of the affected regions with a history of unequal job opportunities [45].

Due to the scarcity of resources and opportunities to make a living, rural Mixteca will experience persistent inequity in education under our synopsis 1. The marginalized population will have no choice but to give up on education and devote their time to seeking an income to sustain themselves, or to emigrate. As a result, the level of schooling under synopsis 1 is expected to be low, under five years (state C1), which is not enough to complete basic education.

Migratory flows will be a pressing issue reflecting the lack of economic growth, and thus, low prospects for decent jobs. However, cluster 1 allows for two different situations: emigrants will either have a strong bond to their region (state A2; scenario 1) or not (state A3, scenarios 2, 3 and 5). The latter describes the current situation, where there is a strong partial dependence of remittances in the region. Even with no bond to Mixteca, remittances will be an important income source in the region. However, the relevance differs between scenarios 1 and 2, and 3 and 5. The first two scenarios, 1 and 2, see financial transfers as the main income source (state D2); the other two scenarios, 3 and 5, show a combination of remittances with labor income to compensate the minimum wage and support domestic consumption (state D3). The dependence on the remittances is also a reflection of low job earnings (state G3) in the region.

The region will also have higher percentages of deprivation in terms of access to basic services (state E1) such as water, electricity, and drainage that encompass the fulfillment of their social rights. This outcome is also a consequence of the lack of an adequate level of education, which prevents the inhabitants from having the knowledge to exert their rights of access to the basic services coverage.

Given the worsening conditions in the area, formal financial services will be not provided, because the security demands for loans by the formal sectors are too high for the local population, and thus the establishment of a formal banking sector is not profitable. The population may have access to informal financial markets (state O2; scenario 5) or no access to any financial markets (state O3; scenarios 1–3). This depends on the economic situation in the region, which could differ between the scenarios.

The depressed economic and social situation with (mainly male) emigrants leaves no (State S1; scenarios 1–3) or limited human and financial resources (State S3; scenario 5) to empower women, although the necessity is obvious. Women are more likely to be engaged in low-productivity activities and work in the informal sector or in unpaid family jobs, and less likely to move to the formal sector compared to men; therefore, empowerment among women will be highly limited. Scenario 5 in this cluster is the only one which envisions a partial empowerment and women's participation in decision-making processes (state S3), probably related to an attempt to move to a higher level of full empowerment.

In synopsis 1, most scenarios are inclined to the pluricultural identity (state B3) obtained through the interaction of two or more communities inside the national territory, or outside of it as a consequence of the migratory flow [74]. The pluricultural identity is also promoted by the interaction of communities in the vicinity during the early years of children's education. Only one scenario, scenario 1, reflects the high ethnic identification bond that permeates through returning emigrants, as well as through those who keep a permanent bond with their ethnicity (state B2). This sense of belonging to an ethnic group promotes community support, mostly on the labor force. Ethnic identity is very

much linked to emigration patterns, but it also maintains a relationship with income sources; while the pluricultural status in two scenarios (3 and 5) generates income from remittances and labor, the other two scenarios depend mostly on remittances due to meager job opportunities and low earnings.

In summary, cluster 1 envisions a path of increased poverty, lack of opportunity for development, persistent emigration, and a society's lack of hope regarding its own future. An energy transition is expected to bring neither success nor better quality of life in Mixteca under this synopsis.

4.4. Cluster 2, Synopsis: "Hope for a Better Future"

A future with low uncertainty regarding governance (state H1) describes a situation in which the government will have developed the capacity to exert effective and efficient decisions, ensuring a proper and informed process as well as stakeholder involvement, hence decreasing the risk of uncertainty among the population. Policies will be open and transparently handled, offering the communities in Mixteca an understanding of the local decisions taken. It will also maintain their focus and address issues in an effort to avoid stagnation and provide certainty regarding the local government commitments, regardless of political administration change. Therefore, low uncertainty (state H1) about future government decisions and potential for economic growth would lead the way to a better future in the region.

A priority of the government will be to eradicate or diminish corruption through an enforced legal system (state P1). To accomplish this objective, on all governmental scales, authorities will address effective transparency and accountability procedures to reach a convincing law enforcement.

In this cluster of scenarios, the energy transition plays a structural role compared to cluster 1, in achieving Mixteca region's potential. To make use of the principally good conditions of high irradiation levels for installing photovoltaic systems (PVs), the energy policy will provide a system of supportive schemes, such as allowing clean energy preferred access to the national grid, subsidizing investments of PV infrastructure, guaranteeing selling prices, capacity building regarding the generation, controlling, maintaining, and marketing of renewable energy, and reinstating energy auctions (state I2). The encouraging legal and economic conditions promote the investment in PVs, leading to lower energy costs and better availability of electricity. Its generation capacity will be more competitive than gas and coal by a significant margin, which will increase its attractiveness as energy storage solutions become prevalent. The good conditions will attract two types of investors with different aims. The primary aim of one type of investor is to provide affordable electricity to the industry clusters in Puebla states and beyond, with less interest in supplying to the region. With the transition of the Mexican economy and a globally shrinking oil demand for Mexican crude oil [75], the new solar-harvesting alternative would provide support as an alternative source of revenue. The second group of investors are locals. Due to trust in the government and good general economic conditions, the inhabitants will start to invest in PVs, with the aim of improving their own supply of electricity, and potentially the competitiveness of their local industry. Supply to nearby communities with larger populations will be possible, although not in the focus of these investors. The momentum of the second group of investors depends largely on the acceptance of the population regarding renewable energy plans and participation (descriptor F). This ranges from a willingness to contribute economically (state F3) to providing "only" labor support (state F2). The good conditions will also lead to a high level of investment in renewable energy development (state J2).

The development of the electricity system will be accompanied by a positive value added (state N2). Due to the high radiation levels combined with a supportive economic environment, investments in PV technology will provide value added to the region, which will impact positively on innovation efforts or education [76]. A high share of the investors

are local; therefore, the value added will stay in the region, fostering local economic growth [77].

The regional energy system in Mixteca would most likely exert a positive contribution on the national and global effort to mitigate climate change (state Q2). The magnitude could lead to a noteworthy decline in greenhouse gas emissions (GHG) beyond the national target of 25% that studies forecast for 2030 [78,79]. Nevertheless, the situation of the climate in the region will be dominated by efforts outside the region. Consequently, the environmental influence is expected to be handled without greater impact on the population (state R2), and the communities will be better prepared to implement mitigation measures, such as adaptations in agricultural practices, or the construction of houses in secured areas away from riverbanks or cliffs.

The positive impacts of the transformation on the region will also promote considerable partnerships between private investors, government, and NGOs, who would support development projects in renewable transition (state M3).

The broad positive economic circumstances, fueled by the energy transformation, will affect the labor markets, i.e., decent jobs with wage rates above the minimum (state G2). This would lead to an increasing relevance of labor income to total income, through which purchasing power is promoted over time; better working conditions will be provided along with stable jobs (state D1). Under better working conditions, it is likely that emigrants will decide to stay in the region, or even return from abroad (state A1). For some of them, working outside the country has provided the capital and skills to start small businesses, reflected in higher rates of self-employment upon their return to Mexico, compared to those with no migration history. Better working conditions in Mexico would bring an opportunity where emigrants do not return to the United States, reincorporating them into the economically active population. This development enforces the relevance of labor income as the main income source.

With a higher and more reliable income, investments in schooling and infrastructure will gain importance. A longer schooling time (state C2) will not only mean building a skilled workforce, but also training future generations in raising awareness of sustainable development, as well as changing the population's attitudes in everyday life. In particular, investments in PVs by local investors will increase the availability of electricity, which will also be used for productive purposes, in contrast to the current situation in Mixteca. According to the statistics [47], most of rural Mixteca is connected to the grid, but availability for productive uses is rather limited. This will be accompanied by more investment in other basic infrastructures, because the financial situation of the communities, as well as the organization of the communities, is improving. However, it is only in scenario 8 that all basic services are available (state E4), and the population is willing and able to support community building with financial resources (state T3). In scenarios 4, 6 and 7, the access to infrastructure is limited (state E3), and the contribution of the inhabitants to community building focuses on labor (state T2). The difference between both types of participation could lie in the degree of income. As long as the population is able to satisfy its basic needs, people will likely provide financing for the area.

Despite the positive economic situation, this will not overturn the impediments to accessing formal financial markets; informal organized credit suppliers will dominate the local financial market. The financial market in Mixteca will still be based on informal banking (state O2). Unlike cluster 1, where loans were mainly used to cover very basic needs, in cluster 2, a switch in the application of the loans to more productive uses to improve the standard of living is likely. Savings will be used to buy assets: farm animals, land, or build an additional room or an improvement to an existing part of the house, such as the roof or a wall. Financial inclusion will remain a challenge for rural communities in the future.

The strong commitment of women and the creation of women-to-women networks along the value chain and decision-making process are vital for the integration of Mixteca's new energy technology at the community level, and ensuring the long-term use of these technologies [80]. These findings reflect women's essential roles as decision-makers, not only in the household but also in their communities. Women use their social network of relatives and friends to introduce products into their communities; thus, they have become trusted advisors, as with household energy. This empowerment reveals the need to involve women in energy projects and the need to incorporate gender into policies on energy transition [80–82]. Although empowerment is foreseen in all scenarios of this cluster, the intensity differs between scenarios 4 and 8, and scenarios 6 and 7. Scenarios 6 and 7 see a partial empowerment (state S3); scenarios 4 and 8 a full achievement (state S2).

All four scenarios of cluster 2 present a future with high ethnic identity (state B2). Ethnicity will be an important quality of the future communities in Mixteca. Returning emigrants will have a sense of belonging to their communities of origin through it. Community organization is inter-related through labor, which, in turn, is promoted by the ethnic bond of the community itself.

In summary, cluster 2 envisions a path of a higher degree of economic development and better quality of life in Mixteca, because of a renewable energy transition. A sense of prosperity in the area derived from stable jobs and earnings, as well as a perception of security regarding the energy transformation, leads to the construction of a better and sustainable future. However, of the derived scenarios, scenario 8 differs from the other scenarios, in particular regarding basic services, women's empowerment, and community organization.

4.5. Comparative Summary

Both clusters show rather divergent developments, and thus distinct future situations. The reason lies in the different state of the main drivers, i.e., those with a large active or otherwise relevant position, and those which are highly interwoven. The main drivers are *P. Legal system* (P2 vs. P1), *H. Governance uncertainties* (H3 vs. H1), and *Q. Climate change* (Q1 vs. Q2). The highly interwoven descriptors are *I. Governmental policies for integrated energy system* (I1 vs. I2), *N. Added Value creation from the renewable energy sector* (N1 vs. N2), *C. Education* (C1 vs. C2), *J. Investments in energy research* (J1 vs. J2), and *M. Cooperation between government, private investors, NGOs* (M1 vs. M3).

The clear separation between both clusters is partly broken up by five of the descriptorstates, which overlap both clusters. These relate to the descriptors *B. Ethnic identification* (state B2), *F. Population acceptance of renewable energy plans and participation* (state F2), *O. Financial market in rural economy* (state O2), *S. Women's empowerment* (state S3), and *T. Community organization* (state T2). However, these descriptors show passive positions in the system or low active positions (Figure 6). Their impacts on development do not exert a compelling influence; rather, they mostly receive the effects from the other descriptors. The changes to these passive descriptors may not necessarily reflect a change in the trend; the change is reflected within the cluster but stays under the same pathway.

5. Discussion

The divergent patterns of both clusters indicate the complexity required to set the transformation process of the region in motion—in both directions. A combination of multiple (relevant) descriptors is necessary for change. The picture may be explained through the concept of transformability brought by Walker et al. [83], as one of the attributes of social–ecological systems (SESs). Walker states that a new system will be established when ecological, economic, political, or social conditions make the existing system implausible. Societal groups may find themselves trapped in an undesirable situation or development process, i.e., in a "basin of attraction", which is wide and deep. Small movements into a new configuration within the same basin are possible, but the outcome of the reconfiguration is not seen by the society as an improvement. An improvement which leads to a new basin would require a large reconfiguration of the descriptors, which would define a new system with new states. Assuming, for the sake of argument, the situation described by cluster

1 as the starting point of an undesirable situation. Only a complete change of the most relevant descriptors would lead to the situation of cluster 2.

Some SESs persist in states where the society cannot meet the basic needs of human well-being or when the societal, environmental, and political factors are degraded to an imminent loss of well-being; Folke refers to them as "dysfunctional states" [84]. In our case, Mixteca seems to move from a current poverty situation to a worsening state in cluster 1, where extreme poverty has persisted for extended periods. These systems may lack the adaptive capacity to reorganize. To escape from the poverty trap, Folke suggests financial and/or political support, external supporting organizations (NGOs), and local developments of innovation. These supporting components are present in cluster 2 in the following states: G2 stable job above minimum wage, H1 low uncertainties in governance, I2 supportive policies on new energy systems, M3 excellent cooperation between government, private *investors and NGOs, P1 law enforcement;* hence, the future of this cluster seems to alleviate poverty. As Folke points out [84], transformational change involves shifts in perception and meaning, societal network configurations, patterns of interactions among different actors, power relations—not only political—and organizational and institutional arrangements. Transformations make use of crises as windows of opportunity and navigate societal transition from a regime in one stability landscape to another. Transformation involves novelty and innovation [85]. It is this window of opportunity of which Folke speaks, in which the energy transformation, as our results suggest for Mixteca, could serve as the means of a major social reconfiguration [19] to improve well-being in the area [17].

Although Walker's concept is used for a static view of possible futures, principally, the concept of basins of attraction could also be useful to explain possible developments of states over time within each cluster, i.e., to derive possible trajectories. A trajectory would describe possible switches from one state to another, and in the longer term to a possible final state. For example, with an improving economic situation over time, the opportunities to participate in installing a local community-based energy system—descriptor *F. Population acceptance of renewable energy plans and participation*—could change. As long the economic situation is comparably bad, interested inhabitants would participate by offering their labor skills (state F2; scenario 6). If the income situation is enhanced, the inhabitants will provide financial resources (state F3; scenarios 4, 7–8). Such perturbations could occur continuously, thus leading to different complex trajectories. However, because the system evolves over time, the system could shift from the domain of influence of one basin of attraction to another, until it reaches a stable landscape [85].

The present study focused on the situation in Mixteca. Thus, an unconditional transfer of the findings to other region is not recommendable, because the social–cultural–economic setting, as well as the climatic conditions, could be different, potentially influencing the findings. However, the presented approach should provide insights as to which descriptors could be relevant for scrutinizing comparable regions, i.e., poverty-ridden areas.

In the center of the analysis was the modelling of the inter-relationship between the descriptor-states, which was based on extended literature research and experts' judgements. The perceptions and knowledge of the experts had some impact on the construction of the Cross-Impact Matrix, and thus on the modelled driving forces of the system. This challenge of potentially, but unintended or even not detected, biased findings is inherent in the CIB method. To reduce the relevance of this challenge the CIB approach builds on the widely accepted criteria for scenario-building, fulfilling plausibility, consistency, traceability, and transparency [7].

6. Conclusions

The transition from fossil fuel-driven energy systems to renewables-based systems brings a suitable opportunity not only for migrating technologies per se, but to reposition political and social dynamics through the configuration of the socio-technical system. New technology implementation as a driver of the creative destruction of old regimes, as pointed out by Geels [19], is necessary to create opportunities for the more widespread dissemination of renewable systems. This new perspective would bring the opportunity to better understand how to enhance the potential for a successful transition by including civil society in the initiatives of transformation, using social interactions in the new configuration. This perspective should also include the analysis of weaknesses and inter-relationships among the actors of the previous regime, which contribute to retaining it as a dominant actor. The shift to the new model of innovation promotes social interaction in the energy system, despite resistance to adopting and disseminating the new and promising configuration. The adoption of this model could be useful in the attempt to establish the pathway to energy transition as the means to alleviate poverty in rural Mixteca. The restructuring of the energy system could provide the opportunity to a just transition among sectors and institutions.

To achieve the transition, small-scale technological innovations should be available for those rural communities where consumers could become producers with their own power installations [86]. This could foster the development of a social network related to energy, from the individual to the community level [87,88]. The potential of the effectiveness and impacts of these motivations in the creation of a sustainable local energy community could lead to de-carbonization, decreasing emigration, self-employment [89], and self-sufficiency of energy supply [88], creating a sustainable environment in the area and a promising energy transition, while attempting to instill less invasive power dynamics among actors.

The presented study emphasizes the relevance of analyzing the societal aspects affecting and being impacted in order to understand a renewable transition. This analysis is required to anticipate outcomes and adapt to undesirable consequences, as shown in cluster 1, by exploring possibilities to advance a desired just transformation to the benefit of the population, as reflected in cluster 2, with the aim of improving living conditions in Mixteca.

The use of the Cross-Impact Balance methodology proved useful in foreseeing several plausible scenarios and conditions under which an energy system in rural Mixteca could be developed. The outcomes of qualitative impacts, such as social, political, cultural, and environmental aspects, helped unfold the probable future storylines of the area. This approach seeks to introduce the contextual elements that make actors reshape their actions to promote a specific path. Our perspective highlights that pertaining actors' engagement in the practical context of the energy transition should be focused on systemic change [19,90]. The change is no longer questioned; rather, the overall direction of such change, as seen from the distinct results presented above.

Through the analysis of the main drivers, it was possible to envisage the relationships among the contextual conditions, and how each of the descriptors influences the multiple reciprocal interactions. The main drivers call for continuous change and adaptation; systematic transition and evaluation can help reflect on how deliberate actions interact in the societal transition dynamics [90]. These reflections, focused on dynamics and processes, can help re-orient interventions and identify new opportunities. CIB has also proved to be an operational tool to foresee the influence and direction of a renewable transition, and how socio-technical arrangements could be simulated. New pathways embodied in new practices or new technology can be envisioned, concretized by the specific conditions in Mixteca. This way, societal embedding in the energy transition and adoption of a new technology can be predicted.

This case has proven useful to show that the identified societal drivers allow the technological transformation to be triggered which, under suitable conditions, could improve living standards and decrease poverty in rural Mixteca. A challenge for the rural area is societal adaptability to the variable contextual conditions, and to the interdisciplinary exchange on the path to a transition. A critical reflexive evaluation of the outcome from the scenario evaluation could bridge the gap between transition dynamics and policies in the developing context. Providing energy access to alleviate poverty is more about understanding the roles that energy can play in the population's daily activities and supporting them to improve their well-being, rather than a shift of technology. If the

transition can be kept open and remain focused on the goal of poverty alleviation, the innovation capacity could contribute to the societal upgrade. This is an opportunity to make a shift in rural communities.

Supplementary Materials: The following are available online at https://www.mdpi.com/article/10 .3390/en14092596/s1, Table S1: Descriptors and states.

Author Contributions: Conceptualization, L.-P.O.-T. and W.-R.P.; methodology, L.-P.O.-T., D.E.F. and W.-R.P.; software use, L.-P.O.-T.; validation, L.-P.O.-T., D.E.F. and W.-R.P.; formal analysis, L.-P.O.-T., D.E.F. and W.-R.P.; investigation, L.-P.O.-T.; resources, L.-P.O.-T. and D.E.F.; data curation, W.-R.P.; writing—original draft preparation, L.-P.O.-T.; writing—review and editing, L.-P.O.-T., D.E.F. and W.-R.P.; supervision, W.-R.P. All authors have read and agreed to the published version of the manuscript.

Funding: This research is funded by CONACYT and DAAD, grant "Länderbezogenes Kooperationsprogramm mit Mexiko: CONACYT-DAAD, 2018". The field trip to Mixteca, Mexico was funded by KIT-ITAS.

Data Availability Statement: The data presented in this paper are available in the article itself, the Appendix A and the Supplementary Material of the article.

Acknowledgments: The authors thank Armin Grunwald, Director of KIT-ITAS, for the financial support for the field trip. L.-P.O.-T. thanks CONACYT-DAAD for the grant received. L.-P.O.-T. also thanks the researchers Enrique Quiroga-González; Karla Cedano; Aarón Sánchez; M. Sandra Caballero; M. Lishey Lavariega; Carlos Menéndez; Alma Soto; Raúl Hernández Garcíadiego, Fernando Manzo, Ing. Álvaro Aguilar Ayón, Mtro. Oscar Andrade; the NGO deputies Anallely Jurado and Valentín Vázquez who interviewed during the field trip and whose insights provided valuable input to this research, as well as providing their valuable time, discussion, and support to this project. The special support from M.Sc. Abel Fragoso, Silvia Sotomayor, Marco A. Castillo and the Spanish—Fregenal—branch deputies from the NGO "Ayuda en Acción" who were kind and willing to provide direct integration into the Mixteca communities, as well as logistic and economic support during the time spent in the area. The authors also wish to thank three anonymous reviewers for their productive comments.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

	AEmigr	B.Eth.id	C.Educ	D.Src inc.	E.Bas.serv	F.REpop	G.Job&earn	.Gov.unce	.PoIES	EnR&I	M.3coop	. AVRe	O.Fin.mkt	P.Leg.sys	Cichg Env-	oc S.Wemp	T.CoOrg
	A1retmig A2permw/b A3permw/o	B1low B2high B3pluri	C1< 5yrs C2-5-9 C3>9yrs	D 1labour D 2remitt D 3lab+rem	E 1none E 2prt-wtr E 3prt-elect E 4all	F1noself F2wk F3fin.res	G1stbel.min G2stabv.mir G3nosec-be G4nosec+at	H 1Iow H 2strw/grw H 3strw/out	11(-) 12(+)	J 11cw J 2high	M1low M2+ M3++	N 1 no N 2yes	O1Itd.form O2acc.inf. O3none	P1law enf. P2aggrav P3current	Q1high Q2low R1high	S 1no S 2 full S 3 part	T1noself T2wk T3fin.res
A Emigration A1 return emigration A2 perm w/bond A3 perm w/o bond B. Ethnic identity		-1 2 2 -1 2 1 2 -1 2	-1 1 -1 2 2 -2 -1 1 -2	1 -1 -1 0 2 2 0 -1 -1	0 0 0 0 -2 1 1 1 0 0 0 0	0 1 1 0 -1 2 0 -1 -1	1 1 1 1 1 -1 1 -1 1 -1 1 -1	-1 1 1 -1 1 1 -1 1 1	0 0 0 0 0 0	0 0 0 0 0 0	-1 1 1 1 -1 -1 1 -1 -1	1 -1 1 -1 1 -1	1 2 0 0 1 0 0 0 0	0 0 0 0 0 0 0 0 0	0 0 0 ¹ 0 0 -1 ¹ 0 0 -1 ¹	-1 1 1 -2 2 1 -2 2 2	0 0 0 0 0 0 0 0 0
B1 low B2 high B3 pluricultural C. Education	1 1 2 1 -1 -2 2 1 2		-1 1 1 1 1 -1 -1 2 1	1 1 1 2 -1 -1 1 2 2	-1 2 2 1 1 1 1 1 -1 -1 1 1 2	0 0 0 0 0 0 0 0 0	-1 1 -1 1 2 -2 1 -2 -1 2 -1 2	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0	0 0 0 0 0 0	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0	0 1 -1 -2 -1 2 1 1 -1	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0	1 0 -1 -2 0 2 -1 0 1	2 -2 0 -2 2 0 -1 1 0
C1 less than 5 years C2 from 5 to 9 years C3 more than 9 years D. Source income	1 2 2 2 1 1 -1 2 2	0 0 0 0 0 1 2 -1 2		1 2 1 1 1 1 2 0 0	2 1 1 -1 -1 2 2 0 0 2 2 2	0 1 -1 0 1 1 0 2 2	2 -2 2 -1 1 0 1 0 -1 2 -1 1	0 0 0 0 0 0 0 0 0	1 -1 1 -1 -1 1	1 -1 0 0 -2 2	0 0 0 -1 1 2 0 0 0	2 -2 -2 2 -3 3	-3 0 2 0 2 -2 2 2 -2	-2 2 0 1 0 1 2 -2 0	2 0 2 - 0 0 -1 - 0 0 -2 2	2 1 -1 -1 -1 -1 1 -1 2 1	0 0 0 0 0 0 -1 0 1
D1 labour D2 remittances D3 remittances+labour E. Basic services access	2 -1 1 1 2 2 1 -1 1 (water, ele	-1 2 0 0 0 1 0 0 1 ctricity, drain	1 1 -3 1 1 -2 1 1 -1 hage)		-1 1 2 1 -1 1 2 1 -1 1 2 2	0 1 1 0 1 2 0 2 2	2 0 2 0 1 1 1 1 0 2 0 2	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0	0 0 0 0 0 0	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0	0 2 -2 1 2 -2 1 2 -2	0 0 0 0 0 0 0 0 0) -2 2 1 -1 -1 1) -1 1 1	0 0 0 0 0 0 0 0 0
E1 none E2 partial w/water E3 partial w/electricity E4 all services	1 2 2 1 1 1 1 1 1 -1 -1 -1	0 0 0 0 0 0 0 0 0 0 0 0	2 -1 -2 1 -1 -1 1 1 -1 -2 2 1	-2 0 0 1 0 0 2 0 0 2 0 0		0 1 -2 0 1 1 0 1 1 0 0 2	1 -1 2 -1 1 -1 1 -1 1 -1 1 -1 -1 2 -1 2	0 0 0 0 0 0 0 0 0 0 0 0	-2 2 -2 2 -2 2 1 -1	-1 1 -1 1 -1 1 1 -1	-1 1 1 -1 1 1 -1 1 1 -1 1 1 1 -1 -1	-2 2 -2 2 -2 2 1 -1	-3 -2 3 -1 1 -1 -1 1 -1 1 2 -2	0 0 0 0 0 0 0 0 0 0 0 0	0 0 -1 · 0 0 2 - 0 0 0 0 (1 0 1 -	-1 0 1 2 0 0 0 0 0 0 1 0 0 0	-2 2 0 -2 2 0 -2 2 0 0 1 1
F. Population acceptance F1 poor participation F2 limited to labour F3 economic contribution	0 0 0 -1 -1 -1 1 1 0	0 0 0 0 0 0 0 0 0 0 0 0	Ians and pa 0 0 0 -1 1 2 -1 2 1	rticipation 0 0 0 1 -1 -1 1 1 1	2 1 1 -2 -1 1 1 1 -2 2 2 2 2		0 0 0 0 1 1 -1 1 1 1 -1 1	0 0 0 0 0 0 0 0 0	0 0 -1 1 -2 2	0 0 -1 1 -2 2	2 -1 -2 -1 1 1 -2 2 2	2 -2 -2 2 -2 2	-1 -1 1 0 1 -1 1 1 -1	0 0 0 0 0 0 0 0 0	1 -1 3 - -1 1 -1 -1 - -1 1 -2 1	3 0 0 0 0 0 0 9 0 0 0	1 -1 -1 -1 1 0 -1 0 1
G1 stab job belw min G2 stab job ab.min G3 no sec.job&belw G4 no sec.job abv min	1 2 2 -2 -2 -2 3 3 3 -1 2 2	0 0 0 0 0 0 0 0 0 0 0 0	1 2 -1 -1 2 2 2 -2 -3 1 2 -1	2 2 2 2 -1 0 0 2 2 2 -2 0	1 1 1 -1 -1 1 1 1 2 -1 -1 -2 -1 1 1 1	0 2 -2 0 1 1 0 1 -2 0 1 1		0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0	-1 1 0 0 -1 1 0 0	-1 1 1 1 -1 -1 -1 1 1 1 -1 -1	0 0 0 0 0 0 0 0	-2 1 -1 2 1 -1 -2 -1 2 -1 1 -1	0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2 -2 -2 -2 2 1 2 -2 -2 -1 2 1	0 2 -1 0 0 1 0 2 -1 0 0 1
H. Governance uncertain H1 low uncertainties H2 strong uncert.w/grw H3 strong unc. w/o grw	11105 -1 -1 -1 -1 1 1 2 1 1 for integral	0 0 0 0 0 0 0 0 0	-2 2 1 -1 2 2 2 -1 -2	2 1 1 1 0 0 -2 2 2	-1 1 1 1 -1 1 2 1 1 -1 -1 -2	-1 1 1 1 1 -1 2 1 -2	2 2 -1 1 1 1 1 2 -1 -2 2 -2		-2 2 -1 1 2 -2	-1 1 1 -1 2 -2	-2 1 2 1 -1 -1 2 -1 -2	-2 2 1 -1 2 -2	1 1 -2 -1 2 -1 -2 -1 2	2 -2 1 1 -1 1 -2 2 1	0 0 -1 ⁻ 1 -1 1 - 1 -1 2 -	0 0 0 1 -2 2 2 2 -2 2 2	0 0 0 -1 1 1 -1 1 -1
I1 restrictive I2 supportive J. Investments in energy J1 low invest, or none	1 2 2 -1 -2 -2 research		1 2 -2 -2 2 2 1 1 -2		2 -1 -1 -1 -2 2 2 2 2 2 1 1 -2	1 -1 -1 -1 1 1 1 -1 -2		-2 1 1 2 1 -1	11-1	3 -3 -3 3	1 -1 -1 -1 1 1 2 -1 -2	2 -2 2	-1 1 -1 1 1 -2		2 -2 3 - -2 2 -3 3 2 -2 3 -		
J2 high investment M. Cooperation between M1 no/low M2 good	0 0 0 governme 1 1 1 -1 -1 -1	0 0 0 nt, private in 0 0 0	-2 1 2 vestors, NG 1 1 -1 -1 1 1	1 -2 0 60s -1 1 1 1 -1 -1	-2 1 1 2 2 -1 -1 -2 -1 1 1 1	-1 1 2 2 -1 -1 -1 1 1	-1 2 -1 1 1 -1 2 -1 -1 1 -1 1		-1 1 2 -2 -2 2	2 -2	-2 1 2	-2 2 2 -2 -1 1	0 0 0 -1 2 -1 1 1 -1	0 0 0 -2 2 1 1 -1 -1	-2 2 -3 : 1 -1 2 -	2 2 -2 -1 -1 1 1	000
M3 excellent N. Added Value creation N1 no/low N2 yes/good	-2 -2 -2 from the re 0 0 0 -1 -1 -1	0 0 0 newable ene 0 0 0 0 0 0	-2 2 2 rgy sector 0 0 -2 2 2	2 -2 -1 -1 0 0 2 0 0	-2 1 1 2 1 -2 -2 -2 -2 2 2 2 2	-2 1 2 2 -2 -2 -2 2 2	-1 2 -2 1 1 -1 1 -1 -1 2 -1 1	2 1 -2 -1 0 0 1 0 0	-3 3 2 -3 -3 3	-3 3 3 -3 -3 3	2 -1 -2	-2 2	2 -1 -2 0 0 0 0 0 0	2 -2 -1 0 0 0 1 0 0	-2 2 -2 1 1 -1 2 - -1 1 -2 1		-1 0 1 0 0 0 -1 1 1
D. Financial market in run O1 ltd access to formal O2 access to informal O3 no access to any	al econom -1 -1 -1 0 0 0 1 2 2		0 1 -1 1 1 -1 2 1 -2	1 2 2 2 1 2 2 2 2 2	-1 1 1 1 -1 1 1 -1 2 -1 -1 -1	1 1 1 1 1 1 -1 2 -2	1 1 -1 1 1 -1 1 -1 1 -2 1 -1	1 0 -1 0 1 1 0 0 0	-2 2 0 0 1 -1	0 0 0 0 0 0	2 1 2 2 1 -1 1 -1 -2	0 0 0 0 0 0		0 0 0 0 0 0 0 0 0	0 0 -1 ⁻ 0 0 0 (0 0 1 -	-2 2 2 -2 2 2 1 2 -2 -2	-1 0 1 -1 1 1 -1 1 -2
P. Legal System P1 law enforcement P2 aggravate P3 not effectively enford	-2 -1 -1 2 2 2 1 1 1	0 0 0 2 -1 0 0 0 0	2 3 1 3 2 -3 1 2 -2	2 -1 -1 1 2 2 1 1 2	-2 2 2 2 2 1 1 -2 1 1 2 -1	-1 1 1 1 1 -1 0 1 -1	1 2 -2 2 2 -3 3 -1 1 -2 2 -1	2 -1 -2 -2 -1 2 -1 1 2	-1 1 1 -1 1 -1	-1 1 2 -2 1 -1	-1 2 3 3 -2 -3 2 -1 -1	-2 2 2 -2 1 -1	2 1 -2 -2 2 2 -1 1 0		0 0 -2 2 0 0 1 - 0 0 1 -	-2 2 1 1 2 -2 -1 1 1 -1 -1	0 0 0 -1 2 -1 -1 1 -1
Q1 high impact Q2 low impact Q2 low impact R. Environmental effects	1 2 2 0 0 0 on populat	0 0 0 0 0 0			222-2	-2 2 1 0 0 0	1 -1 2 -2 -1 1 -2 2		-2 2 0 0	-2 2 0 0	-1 1 1 2 -1 -1	-2 2 0 0			3 - -3 :		
R2 low impact S. Women's empowermen S1 limited or none S2 full attained	nt 0 0 0 -1 -1 0			-20-1	$\begin{array}{c c} -2 & -1 & -2 \\ \hline -2 & -1 & -1 & 2 \\ \hline 0 & 0 & 0 & 0 \\ \hline 0 & 0 & 0 & 2 \end{array}$							00	-1 -1 2 1 2 -2				
S3 partial T. Community organization T1 poor organization T2 limited to labour	0 0 0 n 1 1 1 -1 -1 -1	2 -2 1 -1 2 1	0 0 0 1 -1 0 -1 2 0	-1 1 1 2 -1 -1	2 -1 -1 -2 -1 1 1 0	0 0 1 2 -1 -1 -1 2 -1	0 0 0 0 1 -2 2 -2 2 -1 1 -1	000 -1 -1 1 1 1 -1		000	-1 1 1 2 -1 -1 -1 1 1				2 -2 2 - -1 1 0 0		000
T3 economic contributi	-2 -2 -2	0 0 0	-1 2 0	1 1 1	-2 2 2 2	-1 0 1	1 -1 1 -1	1 1 -1	0 0	0 0	-1 1 1	-1 1	0 1 -1	1 -1 -1	-1 1 -1 '	0 0 0	

Appendix A

Figure A1. Cross-Impact Matrix (CIM).

References

- Scholz, R.W.; Bartelsman, E.J.; Diefenbach, S.; Franke, L.; Grunwald, A.; Helbing, D.; Hill, R.; Hilty, L.; Höjer, M.; Klauser, S. Unintended side effects of the digital transition: European scientists' messages from a proposition-based expert round table. *Sustainability* 2018, 10, 2001. [CrossRef]
- 2. Grunwald, A. Diverging pathways to overcoming the environmental crisis: A critique of eco-modernism from a technology assessment perspective. *J. Clean. Prod.* **2018**, *197*, 1854–1862. [CrossRef]
- 3. Orehounig, K.; Evins, R.; Dorer, V. Integration of decentralized energy systems in neighbourhoods using the energy hub approach. *Appl. Energy* **2015**, 154, 277–289. [CrossRef]
- 4. Østergaard, P.A. Reviewing optimisation criteria for energy systems analyses of renewable energy integration. *Energy* **2009**, *34*, 1236–1245. [CrossRef]
- 5. Vidal-Amaro, J.J.; Østergaard, P.A.; Sheinbaum-Pardo, C. Optimal energy mix for transitioning from fossil fuels to renewable energy sources—The case of the Mexican electricity system. *Appl. Energy* **2015**, *150*, 80–96. [CrossRef]
- 6. Terrapon-Pfaff, J.; Dienst, C.; König, J.; Ortiz, W. A cross-sectional review: Impacts and sustainability of small-scale renewable energy projects in developing countries. *Renew. Sustain. Energy Rev.* **2014**, *40*, 1–10. [CrossRef]
- Dewald, U.; Grunwald, A.; Poganietz, W.-R.; Schippl, J. Die energiewende als sozio-technische transformation von der analyse zur gestaltung. In *Energiewende: Eine Sozialwissenschaftliche Einführung*; Radtke, J., Canzler, W., Eds.; Springer Fachmedien Wiesbaden: Berlin/Heidelberg, Germany, 2019; pp. 319–352. [CrossRef]
- 8. Miller, C.A.; Iles, A.; Jones, C.F. The social dimensions of energy transitions. *Sci. Cult.* **2013**, *22*, 135–148. [CrossRef]

- 9. Geels, F.W. From sectoral systems of innovation to socio-technical systems: Insights about dynamics and change from sociology and institutional theory. *Res. Policy* 2004, *33*, 897–920. [CrossRef]
- Pastukhova, M.; Westphal, K. Governing the global energy transformation. In *The Geopolitics of the Global Energy Transition*; Hafner, M., Tagliapietra, S., Eds.; Springer International Publishing: Berlin/Heidelberg, Germany, 2020; pp. 341–364. [CrossRef]
- 11. Bartiaux, F.; Maretti, M.; Cartone, A.; Biermann, P.; Krasteva, V. Sustainable energy transitions and social inequalities in energy access: A relational comparison of capabilities in three European countries. *Glob. Transit.* **2019**, *1*, 226–240. [CrossRef]
- 12. Bradshaw, M. Global energy dilemmas: A geographical perspective. Geogr. J. 2010, 176, 275–290. [CrossRef]
- 13. Hulme, M. (Still) Disagreeing about climate change: Which way forward. J. Relig. Sci. Spec. Issue 2015, 50, 893–905. [CrossRef]
- 14. Vãran, C.; Crețan, R. Place and the spatial politics of intergenerational remembrance of the Iron Gates displacements in Romania, 1966–1972. *Area* **2018**, *50*, 509–519. [CrossRef]
- 15. Creţan, R.; Vesalon, L. The political economy of hydropower in the communist space: Iron gates revisited. *Tijdschr. Voor Econ. Soc. Geografie* **2017**, *108*, 688–701. [CrossRef]
- 16. Brosemer, K.; Schelly, C.; Gagnon, V.; Arola, K.L.; Pearce, J.M.; Bessette, D.; Schmitt Olabisi, L. The energy crises revealed by COVID: Intersections of Indigeneity, inequity, and health. *Energy Res. Soc. Sci.* **2020**, *68*, 101661. [CrossRef] [PubMed]
- 17. Burke, M.J.; Stephens, J.C. Political power and renewable energy futures: A critical review. *Energy Res. Soc. Sci.* **2018**, *35*, 78–93. [CrossRef]
- 18. Criqui, P.; Mima, S. European climate—energy security nexus: A model based scenario analysis. *Energy Policy* **2012**, *41*, 827–842. [CrossRef]
- 19. Geels, F.W. Regime resistance against low-carbon transitions: Introducing politics and power into the multi-level perspective. *Theory Cult. Soc.* **2014**, *31*, 21–40. [CrossRef]
- 20. Sovacool, B.K. A qualitative factor analysis of renewable energy and Sustainable Energy for All (SE4ALL) in the Asia-Pacific. *Energy Policy* **2013**, *59*, 393–403. [CrossRef]
- Sweeney, S. Working toward energy democracy. In *State of the World 2014: Governing for Sustainability;* Island Press/Center for Resource Economics: Washington, DC, USA, 2014; pp. 215–227. [CrossRef]
- 22. Practical-Action. Poor People's Energy Outlook 2014: Key Messages on Energy for Poverty Alleviation; Practical Action Publishing: Rugby, UK, 2014.
- 23. International-Energy-Agency. World Energy Outlook 2017; IEA: Paris, France, 2017. [CrossRef]
- 24. United-Nations-Development-Programme. *Charting Pathways out of Multidimensional Poverty: Achieving the SDGs*; United-Nations-Development-Programme: New York, NY, USA, 2020; p. 52.
- 25. Singh, P.K.; Chudasama, H. Evaluating poverty alleviation strategies in a developing country. *PLoS ONE* **2020**, *15*, e0227176. [CrossRef] [PubMed]
- 26. Ulsrud, K.; Winther, T.; Palit, D.; Rohracher, H. Village-level solar power in Africa: Accelerating access to electricity services through a socio-technical design in Kenya. *Energy Res. Soc. Sci.* **2015**, *5*, 34–44. [CrossRef]
- 27. Miller, C.A.; Altamirano-Allende, C.; Johnson, N.; Agyemang, M. The social value of mid-scale energy in Africa: Redefining value and redesigning energy to reduce poverty. *Energy Res. Soc. Sci.* 2015, *5*, 67–69. [CrossRef]
- Miller, C.A.; Richter, J.; O'Leary, J. Socio-energy systems design: A policy framework for energy transitions. *Energy Res. Soc. Sci.* 2015, *6*, 29–40. [CrossRef]
- 29. Khan, M.S.; Arefin, T.M.S. Safety net, social protection, and sustainable poverty reduction: A review of the evidences and arguments for developing countries. *IOSR J. Humanit. Soc. Sci.* **2013**, *15*, 23–29. [CrossRef]
- 30. World-Bank. Finance for All? Policies and Pitfalls in Expanding Access; World Bank: Washington, DC, USA, 2008.
- 31. Westover, J. The record of microfinance: The effectiveness/ineffectiveness of microfinance programs as a means of alleviating poverty. *Electron. J. Sociol.* **2008**, *12*, 1–8.
- 32. Montgomery, H.; Weiss, J. Can commercially-oriented microfinance help meet the millennium development goals? Evidence from pakistan. *World Dev.* **2011**, *39*, 87–109. [CrossRef]
- Maisonnave, H.; Pycroft, J.; Saveyn, B.; Ciscar, J.-C. Does climate policy make the EU economy more resilient to oil price rises? A CGE analysis. *Energy Policy* 2012, 47, 172–179. [CrossRef]
- Weimer-Jehle, W.; Buchgeister, J.; Hauser, W.; Kosow, H.; Naegler, T.; Poganietz, W.-R.; Pregger, T.; Prehofer, S.; von Recklinghausen, A.; Schippl, J.; et al. Context scenarios and their usage for the construction of socio-technical energy scenarios. *Energy* 2016, 111, 956–970. [CrossRef]
- 35. Börjeson, L.; Höjer, M.; Dreborg, K.; Ekvall, T.; Finnveden, G. Scenario types and techniques: Towards a user's guide. *Futures* **2013**, *8*, 723–739. [CrossRef]
- Weimer-Jehle, W. Cross-impact balances: A system-theoretical approach to cross-impact analysis. *Technol. Forecast. Soc. Chang.* 2006, 73, 334–361. [CrossRef]
- 37. Weimer-Jehle, W.; Vögele, S.; Hauser, W.; Kosow, H.; Poganietz, W.-R.; Prehofer, S. Socio-technical energy scenarios: State-of-theart and CIB-based approaches. *Clim. Chang.* 2020, *162*, 1723–1741. [CrossRef]
- Schweizer, V.J. Reflections on cross-impact balances, a systematic method constructing global socio-technical scenarios for climate change research. *Clim. Chang.* 2020, *162*, 1705–1722. [CrossRef]
- 39. Büscher, C.; Schippl, J.; Sumpf, P. Introduction. In *Energy as a Sociotechnical Problem: An Interdisciplinary Perspective on Control, Change, and Action in Energy Transitions*, 1st ed.; Routledge: Oxon, UK; New York, NY, USA, 2019; pp. 1–13. [CrossRef]

- 40. Büscher, C.; Ornetzeder, M.; Droste-Franke, B. Amplified socio-technical problems in converging infrastructures: A novel topic for technology assessment. *J. Technol. Assess. Theory Pract.* **2020**, *29*, 11–16. [CrossRef]
- Gobierno-de-Puebla and Secretaría-de-Finanzas-y-Administración. Actualizacion del Programa Regional de Desarrollo 2011–2017; Gobierno-de-Puebla and Secretaría-de-Finanzas-y-Administración: Puebla, Mexico, 2011–2017. Available online: http://www.transparenciafiscal.puebla.gob.mx/index.php?option=com_docman&task=cat_view&gid=581&Itemid=63 (accessed on 5 November 2019).
- 42. Almanza, S.R.; Cajigal, R.E.; Barrientos, A.J. *Actualización de Los Mapas de Irradiación Global Solar en la República Mexicana;* Engineering department of the UNAM: Mexico City, Mexico, 1992.
- 43. Jonathan, S.F. *Diseño y Construcción de un Sistema de Seguimiento Fotovoltaico*; Universidad Tecnológica de la Mixteca: Oaxaca, México, 2012; p. 98.
- 44. Consejo-Nacional-de-Evaluación-de-la-Política-de-Desarrollo-Social-(CONEVAL). Tablas Dinámicas, Medición de la Pobreza; CONEVAL: Mexico City, Mexico, 2018.
- 45. Consejo-Nacional-de-Evaluación-de-la-Política-de-Desarrollo-Social-(CONEVAL). Informe de Evaluación de la Política de Desarrollo Social; CONEVAL: Mexico City, Mexico, 2018.
- 46. Consejo-Nacional-de-Evaluación-de-la-Política-de-Desarrollo-Social-(CONEVAL). Metodología para la medición multidimensional de la pobreza en México; CONEVAL: Mexico City, Mexico, 2019; p. 75.
- INEGI. Censo de Población y Vivienda 2010. Available online: https://www.inegi.org.mx/programas/ccpv/2010/ (accessed on 8 May 2019).
- 48. Centro-de-Estudios-Sociales-y-de-Opinión-Pública. *El Acceso Universal a la Energía Eléctrica;* Cámara de Diputados, LXIII Legislatura: Mexico City, Mexico, 2018.
- 49. Comité-del-centro-de-estudios-de-las-Finanzas-Públicas. *Plan Nacional de Desarrollo 2007–2012*; Cámara-de-Diputados, H., Ed.; LX-Legislatura: Mexico City, Mexico, 2007; p. 91.
- 50. OECD. Improving Education in Mexico: A State level perspective from Puebla; OECD: Paris, France, 2013. [CrossRef]
- 51. Instituto-Nacional-para-la-Evaluación-de-la-Educación. *Breve Panorama Educativo de la Población Indígena;* Edición, A., Ed.; Fondo Editorail INEE: Mexico City, Mexico, 2013.
- 52. Instituto-Nacional-de-Estadistica-y-Geografia-INEGI-. *Anuario Estadistico y Geografico de Puebla 2017;* INEGI: Puebla, Mexico, 2017.
- 53. Fundación-BBVA-Bancomer-A.C.; Consejo-Nacional-de-Población. *Yearbook of Migration and Remittances Mexico* 2019; Gobernación, S.d., Ed.; Surtidora Gráfica, S.A. de C.V.: Mexico City, Mexico, 2019; Volume 7, p. 188.
- 54. OECD. Financial Inclusion in Mexico Remains the Lowest amongst OECD Countries; OECD: Paris, France, 2017. [CrossRef]
- 55. The-World-Bank. Mexico Financial Inclusion Development Policy Financing; The-World-Bank: Washington, DC, USA, 2019.
- 56. Servín, A. Inclusión Financiera Para Zonas Rurales, Desafío Para el Nuevo Gobierno, in El Economista; Nacer Global: Mexico City, Mexico, 2018.
- 57. Pring, C.; Vrushi, J. *Global Corruption Barometer Latin America and the Caribbean 2019*; Transparency International: Berlin, Germany, 2019.
- 58. Vilalta, C. Vote-buying crime reports in Mexico: Magnitude and correlates. Crime Law Soc. Chang. 2010, 54, 325–337. [CrossRef]
- 59. Muno, W.; Faust, J.; Thunert, M. Mexico Report; Bertelsmann Stiftung: Gütersloh, Germany, 2019.
- 60. Rodriguez-Sanchez, J.I. *Measuring Corruption in Mexico*; James, A., Ed.; Baker III Institute for Public Policy of Rice University: Houston, TX, USA, 2018; p. 17.
- 61. Tornel, C.; Gutiérrez, M.; Villarreal, J. *Energy Transition in Mexico: The Social Dimension of Energy and the Politics of Climate Change;* Iniciativa climatica de Mexico: Mexico City, Mexico, 2019.
- 62. Viscidi, L.; Graham, N.; Phillips, S. *Mexican Power Sector Policies. Economic and Trade Impacts*; The Dialogue leadership for the Americas: Washington, DC, USA, 2020.
- Bellini, E.; Zarco, J. Mexican Government Halts Grid Connection of New Solar and Wind Projects. Available online: https://www. pv-magazine.com/2020/05/05/mexican-government-halts-grid-connection-of-new-solar-and-wind-projects/ (accessed on 5 May 2020).
- 64. Meylan, G.; Seidl, R.; Spoerri, A. Transitions of municipal solid waste management. Part I: Scenarios of Swiss waste glasspackaging disposal. *Resour. Conserv. Recycl.* 2013, 74, 8–19. [CrossRef]
- 65. Schütze, M.; Seidel, J.; Chamorro, A.; León, C. Integrated modelling of a megacity water system—The application of a transdisciplinary approach to the Lima metropolitan area. *J. Hydrol.* **2019**, *573*, 983–993. [CrossRef]
- Mowlaei, M.; Talebian, H.; Talebian, S.; Gharari, F.; Mowlaei, Z.; Hassanpour, H. Scenario Building for Iran Short-Time Future (Results of Iran Futures Studies Project). In Proceedings of the Scenario Building for Iran Short-Time Future (Results of Iran Futures Studies Project), Hong Kong, China, 22–24 September 2016.
- 67. Yaghoobi, N.; Dehghani, M.; Omidvar, M. Foresight of entrepreneurial university using the integrated method of processing scenarios and cross-impact analysis 1404. *Product. Manag.* **2018**, *11*, 45–74.
- Hummel, E.; Hoffmann, I. Complexity of nutritional behavior: Capturing and depicting its interrelated factors in a cause-effect model. *Ecol. Food Nutr.* 2016, 55, 241–257. [CrossRef] [PubMed]
- 69. Kurniawan, J. Discovering Alternative Scenarios for Sustainable Urban Transportation. In Proceedings of the 48th Annual Conference of the Urban Affairs Association, Toronto, TO, Canada, 4–7 April 2018.

- Norouzi, N.; Fani, M.; Ziarani, Z.K. The fall of oil Age: A scenario planning approach over the last peak oil of human history by 2040. J. Pet. Sci. Eng. 2020, 188, 106827. [CrossRef]
- 71. Cross-Impact Balances. Available online: http://www.cross-impact.org (accessed on 2 May 2020).
- 72. Transparency-International. *Corruption Perceptions Index 2019;* Transparency-International: Berlin, Germany, 2019; p. 34. ISBN 978-3-96076-134-1.
- 73. Gobierno-del-Estado-de-Puebla. Síntesis de la Estrategia de Mitigación y Adaptación del Estado de Puebla Ante el Cambio Climático; Secretaría-de-Sustentabilidad-Ambiental-y-Ordenamiento-Territorial, Ed.; Gobierno-del-Estado-de-Puebla: Puebla, Mexico, 2011.
- 74. Fox, J. Reframing Mexican migration as a multi-ethnic process. *Lat. Stud.* **2006**, *4*, 39–61. [CrossRef]
- 75. DNV-GL-Group. Oil and gas forecast to 2050. Energy Transition Outlook 2017; DNV GL AS: Hovik, Norway, 2017; p. 76.
- 76. Warneryd, M.; Karltorp, K. The role of values for niche expansion: The case of solar photovoltaics on large buildings in Sweden. *Energy Sustain. Soc.* **2020**, *10*, 7. [CrossRef]
- 77. Appunn, K. Investment in Renewables Creates Added Value and Local Jobs-Study. Available online: https://www. cleanenergywire.org/news/investment-renewables-creates-added-value-and-local-jobs-study (accessed on 15 March 2021).
- 78. Elizondo, A.; Pérez-Cirera, V.; Strapasson, A.; Fernández, J.C.; Cruz-Cano, D. Mexico's low carbon futures: An integrated assessment for energy planning and climate change mitigation by 2050. *Futures* **2017**, *93*, 14–26. [CrossRef]
- 79. International-Energy-Agency. Mexico Energy Outlook 2016; International-Energy-Agency: Paris, France, 2016.
- 80. Heuër, A. Women-to-women entrepreneurial energy networks: A pathway to green energy uptake at the base of pyramid. *Sustain*. *Energy Technol. Assess.* **2017**, *22*, 116–123. [CrossRef]
- 81. Permana, A.S.; Aziz, N.A.; Siong, H.C. Is mom energy efficient? A study of gender, household energy consumption and family decision making in Indonesia. *Energy Res. Soc. Sci.* 2015, *6*, 78–86. [CrossRef]
- Batliwala, S.; Reddy, A.K.N. Energy for women and women for energy (engendering energy and empowering women)1 1A rudimentary version of this paper was presented at the Brainstorming Meeting of ENERGIA: Women and Energy Network on June 4-5, 1996, at the University of Twente, Enschede, the Netherlands (cf. [Batliwala and Reddy, 1996]). *Energy Sustain. Dev.* 2003, 7, 33–43. [CrossRef]
- 83. Walker, B.; Holling, C.S.; Carpenter, S.R.; Kinzig, A. Resilience, adaptability and transformability in social–ecological systems. *Ecol. Soc.* **2004**, *9*, 9. [CrossRef]
- Folke, C.; Chapin, F.S.; Olsson, P. Transformations in ecosystem stewardship. In *Principles of Ecosystem Stewardship: Resilience-Based Natural Resource Management in a Changing World*; Folke, C., Kofinas, G.P., Chapin, F.S., Eds.; Springer: New York, NY, USA, 2009; pp. 103–125. [CrossRef]
- 85. Gallopín, G.C. Linkages between vulnerability, resilience, and adaptive capacity. *Glob. Environ. Chang.* **2006**, *16*, 293–303. [CrossRef]
- 86. Zalengera, C.; Blanchard, R.E.; Eames, P.C. Putting the End-User First: Towards Addressing Contesting Values in Renewable Energy Systems Deployment for Low-Income Households—A Case from Likoma Island, Malawi; Springer International Publishing: Berlin/Heidelberg, Germany, 2015; pp. 101–112.
- 87. Dóci, G.; Vasileiadou, E. Let's do it ourselves. Individual motivations for investing in renewables at community level. *Renew. Sustain. Energy Rev.* **2015**, *49*, 41–50. [CrossRef]
- Cabraal, R.A.; Barnes, D.F.; Agarwal, S.G. Productive uses of energy for rural development. *Annu. Rev. Environ. Resour.* 2005, 30, 117–144. [CrossRef]
- 89. Lowitzsch, J.; Hanke, F. Consumer (Co-)ownership in Renewables, Energy Efficiency and the Fight Against Energy Poverty-a Dilemma of Energy Transitions. *Renew. Energy Law Policy Rev.* 2019, *9*, 5–21. [CrossRef]
- 90. Verbong, G.; Loorbach, D. Civil society in sustainable energy transitions: Adrian Smith. In *Governing the Energy Transition: Reality, Illusion or Necessity*, 1st ed.; Routledge: London, UK; Taylor & Francis Group: New York, NY, USA, 2012; p. 392.